



University of Dundee

Single-inhaler triple therapy in patients with chronic obstructive pulmonary disease

Langham, Sue; Lewis, Jen; Pooley, Nick; Embleton, Nina; Langham, Julia; Han, MeiLan K.

Published in: Respiratory Research

DOI: 10.1186/s12931-019-1213-9

Publication date: 2019

Document Version Publisher's PDF, also known as Version of record

Link to publication in Discovery Research Portal

Citation for published version (APA):

Langham, S., Lewis, J., Pooley, N., Embleton, N., Langham, J., Han, M. K., & Chalmers, J. D. (2019). Singleinhaler triple therapy in patients with chronic obstructive pulmonary disease: a systematic review. *Respiratory Research*, *20*(1), 1-13. [242]. https://doi.org/10.1186/s12931-019-1213-9

General rights

Copyright and moral rights for the publications made accessible in Discovery Research Portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- Users may download and print one copy of any publication from Discovery Research Portal for the purpose of private study or research.
- · You may not further distribute the material or use it for any profit-making activity or commercial gain.
- You may freely distribute the URL identifying the publication in the public portal.

Take down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

REVIEW

Single-inhaler triple therapy in patients with chronic obstructive pulmonary disease: a systematic review

Sue Langham^{1*}, Jen Lewis¹, Nick Pooley¹, Nina Embleton¹, Julia Langham¹, MeiLan K. Han² and James D. Chalmers³

Abstract

Background: Guidelines recommend that treatment with a long-acting β_2 agonist (LABA), a long-acting muscarinic antagonist (LAMA), and inhaled corticosteroids (ICS), i.e. triple therapy, is reserved for a select group of symptomatic patients with chronic obstructive pulmonary disease (COPD) who continue to exacerbate despite treatment with dual therapy (LABA/LAMA). A number of single-inhaler triple therapies are now available and important clinical questions remain over their role in the patient pathway. We compared the efficacy and safety of single-inhaler triple therapy to assess the magnitude of benefit and to identify patients with the best risk-benefit profile for treatment. We also evaluated and compared study designs and population characteristics to assess the strength of the evidence base.

Methods: We conducted a systematic search, from inception to December 2018, of randomised controlled trials (RCTs) of single-inhaler triple therapy in patients with COPD. The primary outcome was the annual rate of moderate and severe exacerbations.

Results: We identified 523 records, of which 15 reports/abstracts from six RCTs were included. Triple therapy resulted in the reduction of the annual rate of moderate or severe exacerbations in the range of 15–52% compared with LAMA/LABA, 15–35% compared to LABA/ICS and 20% compared to LAMA. The patient-based number needed to treat for the moderate or severe exacerbation outcome ranged between approximately 25–50 (preventing one patient from having an event) and the event-based number needed to treat of around 3–11 (preventing one event). The absolute benefit appeared to be greater in patients with higher eosinophil counts or historical frequency of exacerbations and ex-smokers. In the largest study, there was a significantly higher incidence of pneumonia in the triple therapy arm. There were important differences in study designs and populations impacting the interpretation of the results and indicating there would be significant heterogeneity in cross-trial comparisons.

Conclusion: The decision to prescribe triple therapy should consider patient phenotype, magnitude of benefit and increased risk of adverse events. Future research on specific patient phenotype thresholds that can support treatment and funding decisions is now required from well-designed, robust, clinical trials.

Trial registration: PROSPERO #CRD42018102125.

Keywords: Chronic obstructive pulmonary disease, Inhalers, Combination drug therapy, Disease exacerbation

* Correspondence: suelangham@maverex.com

¹Maverex Limited, Manchester, UK



© The Author(s). 2019 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.



Open Access

Full list of author information is available at the end of the article

Background

Chronic obstructive pulmonary disease (COPD) is a leading cause of mortality and morbidity worldwide [1]. Pharmacological treatment relies predominately on inhaled bronchodilators and anti-inflammatory agents [2]. The 2019 Global Initiative for Chronic Obstructive Lung Disease (GOLD) strategy document, based on the bestavailable evidence from the published literature, recommend that choice of treatment depends on symptom and exacerbation severity. In those patients with a high risk of exacerbations, therapy relies on a long-acting muscarinic antagonist (LAMA) or if a patient is highly symptomatic, dual therapy with a LAMA and a longacting β_2 agonist (LABA). Evidence suggests that dual therapy with LABA and inhaled corticosteroids (ICS) and the step up to triple therapy (LAMA/LABA/ICS) be considered for a select group of patients who continue to exacerbate despite appropriate treatment and/or features suggesting steroid responsiveness.

Triple therapy, provided as multiple inhalers, has in pooled analyses been shown to improve lung function, health-related quality of life and exacerbations [3-7]. However, evidence suggests that triple therapy is often over prescribed in clinical practice and used in patients who are not frequent exacerbators [8–10]. Recently, two single-inhaler triple therapies have received marketing authorisation from the European Medicines Agency and one other is in late-stage clinical development [11]. A systematic and critical review of the evidence base for single-inhaler triple therapy is warranted to support clinical decision making for the following reasons. First, important questions in clinical practice remain over the role of triple therapy, which include the magnitude of clinical benefit and the identification of patients with the best risk-benefit profile for treatment. Second, the randomised controlled trial data for single-inhaler therapies will be used to inform the evidence base for triple therapy as a whole, therefore it is important to understand its strengths and limitations. To the best of our knowledge there are currently no systematic reviews that give a comprehensive and critical assessment of the evidence base for single-inhaler triple therapy.

Following the GOLD 2019 update, such a review would provide insights into patients most suited to triple therapy, the strength of the evidence base and the need for future research. The main objective of this study was to conduct a systematic review of randomised controlled trials comparing the efficacy and safety of fixed-dose combinations of a LABA, LAMA and ICS (single-inhaler triple therapy) with LABA, LAMA, LABA/LAMA or LABA/ICS for the treatment of adult patients with COPD. We present the relative and absolute benefit of single-inhaler triple therapy overall and for specific patient subgroups for each trial. We also present estimates of number needed to treat (NNT) and an overview of risks associated with treatment to support clinical decision making. In addition, we present and discuss the characteristics of each trial, highlighting how study design features could impact the interpretation of clinical trial results and cross-trial comparisons.

Methods

We followed the Preferred Reporting Items for Systematic Reviews and Meta Analyses (PRISMA) guidelines for conducting and reporting systematic reviews [12]. The study protocol was prepared and published via PROSPERO (#CRD42018102125) [13]. Eligibility criteria included (see Additional file 1: Table S1): population of adult patients with COPD; intervention of single-inhaler triple therapy; comparators of LABA, LAMA, LAMA/ LABA or LABA/ICS; and study design of parallel-group randomised controlled trials of ≥ 3 months duration. Single-inhaler triple therapies of interest included glycopyrronium bromide/formoterol fumarate/beclomethasone (GLY/FOR/BDP; Trimbow[®]), umeclidinium/vilanterol/fluticasone furoate (UMEC/VI/FF; Trelegy Ellipta), GLY/ FOR/budesonide (GLY/FOR/B; PT010) and GLY/indacaterol/mometasone furoate (GLY/IND/MF; QVM149). The primary outcome of interest was the annual rate of moderate and severe exacerbations. Moderate exacerbations were those that required treatment with systemic corticosteroids and/or antibiotics. Severe exacerbations were those that required hospitalisation or resulted in death [14]. Secondary outcomes included time to first exacerbation, lung function, quality of life and safety outcomes (including serious adverse events [SAEs], pneumonia and mortality).

Searches for full-text reports and conference abstracts containing original data (in English) were run in: MED-LINE, EMBASE, Cochrane Central Register of Controlled Trials, ClinicalTrials.gov and the World Health Organization International Clinical Trial Registry Platform. The final update was in December 2018. The detailed search strategy is available in Additional file 1: Table S2.

Two reviewers (SL and JLe) independently screened the titles and abstracts of citations, and then full-text reports/conference abstracts according to the protocol. Those studies, containing original data, which met the eligibility criteria of this review were included. Disagreements were resolved through consensus or consultation with a third reviewer (JLa or NP). Data were extracted by one author (JLe) and checked independently by two reviewers (SL and NP). Disagreement between the authors was solved by consensus or involvement of a fourth reviewer (JLa). For the primary outcome we extracted data for the overall population and for several phenotypic subgroups including prior exacerbation frequency, blood eosinophil levels/counts and smoking status.

The risk of bias was assessed according to recommendations outlined in the Cochrane Handbook for Systematic Reviews of Interventions, and graded [15]. The tool includes an assessment of sequence generation, concealment of allocation, blinding of participants and investigators, incomplete outcome data and selective outcome reporting. Each source of bias was graded as having high, low or unclear risk.

Data assimilation on the primary and secondary outcomes was through descriptive analysis in the form of tables, figures and descriptive forest plots. Rate ratios (RRs) for the comparison of the annual rate of moderate and severe exacerbations between triple therapy and comparators were displayed, for each study separately, on a forest plot for the overall trial population and for phenotypic subgroups. Where RR and 95% confidence intervals (CIs) were not available from each study they were calculated from the rate of annual moderate or severe exacerbation for triple therapy and comparators and the number of patients per subgroup using the RR in 'R' version 3.3.2. Where RR were available from each study, but 95% CIs were graphically displayed, we approximated the CIs from the forest plots reported in the original publication using graphical software (GetData Graph Digitizer).

We also calculated patient- and event-based NNT for the primary outcome for each of the 12-month studies. Patient-based NNT demonstrates the number of patients that need to be treated with triple therapy relative to the comparator to prevent one patient from having ≥ 1 moderate or severe exacerbation over one year. As the proportion of patients with ≥ 1 exacerbation was rarely reported, proportions were approximated from the Kaplan–Meier curves for the time to the first exacerbation using graphical software (GetData Graph Digitizer). Event-based NNT demonstrates the number of patients that need to be treated to prevent one moderate or severe exacerbation. These were calculated based on the annual rate of moderate or severe exacerbations.

Results

The initial search returned 523 references after deduplication. From these, we identified 105 as potentially relevant. A full-text analysis led to the removal of 90, leaving 15 records (seven abstracts/posters) belonging to six studies for inclusion (Fig. 1) [16–30]. Detailed reasons for exclusion are outlined in Additional file 1: Table S3. The majority of studies were excluded because they reported results for interventions or comparators not included in the eligibility criteria.

The characteristics of the included studies are displayed in Table 1. The six studies included in this review had enrolled 19,658 participants with COPD (range 1810 to 10,355). All of the included studies were multinational, randomised, double-blind, parallel-group trials. Four of the six studies were 1-year duration (TRINITY [16], TRILOGY [17], TRIBUTE [23] and InforMing the PAthway of COPD Treatment [IMPACT] [25]). The primary endpoint in one study was assessed after 24 weeks, but a subset of patients continued treatment for ≤ 1 year (Lung FUnction and quality of LiFe assessment in COPD with closed triple therapy [FULFIL] [24]). The final study was 24-weeks duration (KRONOS [26]).



Study	Treatment	Kev	Study length	Number	FEV ₁ , post-	CAT score	Exacerbation	Exacerbations i	n the previous y	/ear	Prior	Prior F	atients
×		comparisons)	of patients	bronchodilator, % of predicted normal value	≥10 Mean (SD)	rate (range)	≥1 moderate or severe	≥2 moderate or severe	≥1 severe	triple therapy ^c	S	with asthma
TRINITY [16]	GLY/FOR/BDP (Trimbow)	Triple therapy (single inhaler)	52 weeks	2691	36.6%	100% 21.5 (5.8)	1.3 (1–11)	NR	NR	R	Excluded	77% 1	Excluded
	OIT	versus TIO ^a			36.6%	100% 21.6 (5.8)	1.3 (1–5)	R	R	NR		78%	
	FOR/BDP + TIO				36.7%	100% 21.7 (6.0)	1.2 (1–7)	R	R	NR		73%	
TRILOGY [17]	GLY/FOR/BDP (Trimbow)	Triple therapy versus LABA/	52 weeks	1368	36.9%	100% 20.8 (5.9)	1.2 (1–5)	R	NR	NR	Excluded	75% 1	Excluded
	FOR/BDP	S			36.2%	100% 20.8 (5.7)	1.2 (1–6)	R	NR	NR		73%	
TRIBUTE [23]	GLY/FOR/BDP (Trimbow)	Triple therapy versus LAMA/	52 weeks	1532	36.4%	100% NR	1.2 (1–6)	100%	20%	NR	Excluded	66% I	Excluded
	GLY/ND	LABA			36.4%	100% NR	1.2 (1–4)	100%	18%	NR		64%	
Fulfil ^b [24]	UMEC/VI/FF (Trelegy Ellipta)	Triple therapy versus LABA/	24 weeks (extension to 52 weeks)	1810 (430)	47.1%	100% NR	NR	70%	34%	NR	32%	966%	Excluded
	FOR/B + placebo	S			45.4%	100% NR	NR	67%	31%	NR	33%	67%	
IMPACT [25, 31]	UMEC/VI/FF (Trelegy Ellipta)	Triple therapy versus LAMA/	52 weeks	10,355	45.7%	100% 20.1 (6.1)	NR	99.95%	55%	26%	38%	72% 8	Excluded ^d
	UMEC/VI	LABA Triple therapy versus LABA/			45.4%	100% 20.2 (6.2)	NR	%06.66	55%	25%	40%	72%	
	VI/FF	ICS			45.5%	100% 20.1 (6.1)	NR	99.88%	54%	26%	38%	70%	
KRONOS [26]	GLY/FOR/B	Triple therapy versus LAMA/	24 weeks	1902	NR	100% 18.7 (6.4)	0.4 (0.8) ^e 0.0 (0–8) ^f	27%	7%	NR	31%	72% 1	Excluded
	GLY/FOR	LABA Triple therapy versus LABA/			NR	100% 18.1 (6.1)	0.3 (0.7) ^e 0.0 (0–5) ^f	24%	7%	NR	28%	71%	
	FOR/B	ICS			NR	100% 18.4 (6.6)	0.3 (0.6) ^e 0.0 (0–4) ^f	25%	6%	NR	34%	71%	
B budesonide,	BDP beclomethason	e, CAT COPD asses	sment test, COPD chronic	c obstructiv	e pulmonary diseas	se. FEV ₁ forced ex	piratory volume	in one second, FF	fluticasone furoa	te, FOR for	rmoterol fun	narate, F	ULFIL

Table 1 Overview of included studies and study characteristics at haseline

5 μ MING Lung FUnction and quality of LiFe assessment in COPD with closed trlpLe therapy, GLY glycopyrronium bromide, /CS inhaled corticosteroids, *IMPACT* InforM long-acting *B*₂ agonist, *LAMA* long-acting muscarinic antagonist, *MR* not reported, *SD* standard deviation, *TIO* tiotropium, *UMEC* umeclidinium, *VI* vilanterol "Other comparisons were triple therapy (single inhaler) versus LABA/ICS + LAMA and TIO versus LABA/ICS + LAMA and ATO versus LABA/ICS + LAMA and TIO versus LABA/ICS + LAMA and ATO versus LABA/ICS + LAMA and ATO versus LABA/ICS + LAMA and NAAAA and ATO versus LABA/ICS + LAMA and ICS a finduding LAMA, LAMA and ICS a Patients with a prior history of asthma, but not those with a current diagnosis of asthma, were eligible for inclusion fistory of asthma, but not those with a current diagnosis of asthma, were eligible for inclusion fistory of asthma, but not those with a current diagnosis of asthma, were eligible for inclusion fieldian (range) fieldian (range)

The mean age of participants in the different studies was relatively similar (63.3-65.9 years). All the studies had more male than female participants (66-77% males). Where reported, the duration of COPD was similar between five of studies (7.7-8.2 years); the duration of COPD in the KRONOS study was slightly lower (5.4-6.2 years).

Disease-specific study characteristics are outlined in Table 1. Participants' post-bronchodilator forced expiratory volume in one second (FEV₁) was 36% of predicted normal value in three of the studies (TRINITY, TRIL-OGY and TRIBUTE), and 45–47% in two of the studies (FULFIL and IMPACT); the percentage predicted post-bronchodilator FEV₁ was not reported in the KRONOS study. In terms of symptoms, patients from all the studies were required to have a baseline COPD assessment test (CAT) score of \geq 10 (CAT scores range from 0 to 40). The baseline CAT score was 20–21 in the TRIN-ITY, TRILOGY, TRIBUTE and IMPACT studies; and 18–19 in the KRONOS study (the FULFIL study did not report a mean baseline CAT score).

Patients from all studies, except the FULFIL and KRO-NOS studies, were required to have had ≥ 1 exacerbation in the previous year. Patients enrolled into the TRINITY, TRILOGY and TRIBUTE studies all had ≥ 1 moderate or severe exacerbation in the previous 12 months. The FULFIL study inclusion criteria were either $FEV_1 < 50\%$ predicted or ≥ 2 moderate or ≥ 1 severe exacerbation with FEV_1 50-80% predicted; and the IM-PACT study inclusion criteria were either ≥ 1 moderate or severe exacerbation with $FEV_1 < 50\%$ predicted, or ≥ 2 moderate or ≥ 1 severe exacerbation(s) with FEV₁ 50-80% predicted. Patients in the KRONOS study were not required to have had a COPD exacerbation within the previous year. Exacerbation rate was a primary objective for the TRINITY, TRIBUTE and IMPACT studies and a secondary objective for the TRILOGY, FULFIL and KRONOS studies (lung function was the primary endpoint for these studies).

Three of the studies (TRINITY, TRILOGY and TRIB-UTE) excluded patients who were already on triple therapy and did not permit other COPD treatments during the trial; however, 64–78% of patients had received ICScontaining regimens prior to the study start. The FUL-FIL and IMPACT studies included patients who were already on triple therapy and allowed patients to take their existing medications during the run-in period. Around 32–40% of patients were receiving triple therapy during the run-in period and 66–72% of patients were receiving a regimen that included an ICS. In the KRO-NOS study, 23–32% of patients had previously taken triple therapy, and were not permitted to take LAMA, LABA, LAMA/LABA or LABA/ICS during the trial; 71– 73% of patients were receiving ICS at screening. All of the studies excluded patients with a current diagnosis of asthma. Patients with a prior history of asthma were eligible to enrol in the IMPACT study if they had a current diagnosis of COPD.

The studies generally had a low risk of bias according to the Cochrane risk of bias tool [15] (see Additional file 1: Figure S1). There was an unclear risk of bias for selection (allocation concealment), performance bias (blinding of participants and personnel) and detection bias (blinding of outcome assessment) in the FULFIL study due to insufficient information. For the IMPACT study, there was an unclear risk of detection bias (blinding of outcome assessment) also due to insufficient information. The TRINITY, TRILOGY and TRIBUTE studies were funded by Chiesi Farmaceutici, the FULFIL and IMPACT studies were funded by GlaxoSmithKline, and the KRONOS study was funded by Pearl Therapeutics – a member of the AstraZeneca Group.

Primary outcome

The rate of moderate or severe exacerbations during treatment among patients assigned to each intervention is outlined in Fig. 2. The rates were much higher in the IMPACT study and the LAMA/LABA group in the KRONOS study compared with the other studies. Figure 3 outlines the RR, 95% CIs and patient- and event-based NNTs for triple therapy compared with each comparator for each study for the overall population and each subgroup for the primary outcome. The graph is a descriptive display of the data for each study.

Overall population

For overall populations, the rate of moderate or severe exacerbations was generally significantly lower for triple therapy than the comparators. In the studies that assessed single-inhaler triple therapy with a LAMA/LABA (GLY/FOR/BDP versus GLY/IND; UMEC/VI/FF versus UMEC/VI; and GLY/FOR/B versus GLY/FOR), triple therapy resulted in RR of: 0.85 (95% CI: 0.72, 0.995; 15% difference; p = 0.043) [23]; 0.75 (95% CI: 0.70, 0.81; 25% difference; p < 0.001) [25]; and 0.48 (95% CI: 0.37, 0.64; 52% difference; p < 0.001) [26]. The patient-based NNT with triple therapy compared with LAMA/LABA ranged between approximately 33 (IMPACT) and 50 (TRIBUTE). The event-based NNT ranged from 3 (IMPACT) to 11 (TRIBUTE).

In the studies that assessed single-inhaler triple therapy with a LABA/ICS (GLY/FOR/BDP versus FOR/BDP; UMEC/VI/FF versus VI/FF; UMEC/VI/FF versus FOR/B; GLY/FOR/B versus FOR/B), triple therapy resulted in RR of: 0.77 (95% CI: 0.65, 0.92; 23% difference; p = 0.005) [17]; 0.85 (95% CI: 0.80, 0.90; 15% difference; p < 0.001) [25]; 0.65 (95% CI: 0.49, 0.86; 35% difference; p = 0.002) [24] and 0.82 (95% CI: 0.58, 1.17; 18% difference; p = 0.2792) [26].



The patient-based NNT with triple therapy compared to LABA/ICS was around 25. The event-based NNT ranged from 6 (IMPACT) to 8 (TRILOGY). In the study that assessed single-inhaler triple therapy with a LAMA (GLY/FOR/BDP versus tiotropium [TIO]), triple therapy resulted in a RR of 0.80 (95% CI: 0.69, 0.92; 20% difference; p = 0.0025), with a patient-based NNT of around 25 and an event-based NNT of 9.

Prior exacerbation frequency subgroup

For the prior exacerbation frequency subgroup, triple therapy demonstrated a higher reduction in the rate of moderate and severe exacerbations in patients with a history of more exacerbations than those who experienced fewer exacerbations compared to LAMA/LABA (IMPACT [21]), LABA/ICS (TRILOGY) and LAMA (TRINITY). However, the converse was also found in comparisons with LAMA/LABA (TRIBUTE) and LABA/ ICS (IMPACT [21]) where triple therapy demonstrated a higher reduction in the rate of moderate and severe exacerbations in patients who experienced fewer historical exacerbations compared to those with a higher frequency of historical exacerbations. A post-hoc analysis of data from the FULFIL study showed that, compared with LABA/ICS, triple therapy resulted in a significantly higher reduction in moderate and severe exacerbation rates in patients who had experienced ≥ 1 severe exacerbation, and 0/1 moderate exacerbations; patients who experienced ≥ 2 moderate exacerbations also experienced a rate reduction but the difference was not statistically significant [20]. There were no data relating to prior exacerbation frequency subgroups in the KRONOS study.

Eosinophil subgroup

For the eosinophil subgroup, in the TRINITY, TRIB-UTE, IMPACT and KRONOS studies there was a higher reduction in the rate of moderate and severe exacerbations in patients with higher eosinophil counts than those with lower counts for triple therapy comparisons with LAMA/LABA (TRIBUTE, IMPACT and KRONOS) and LAMA (TRINITY). However, in the TRILOGY study (triple therapy [GLY/FOR/BDP] versus LABA/ICS), there was no association between blood eosinophil concentration and the rate of moderate and severe exacerbations. In the IMPACT study there was a higher reduction in the rate of moderate and severe exacerbations in patients with lower eosinophil counts than those with higher counts for triple therapy compared with LABA/ICS. In the KRONOS study (triple therapy [GLY/FOR/B] versus LABA/ICS), there were similar rates of moderate and severe exacerbations across most eosinophil subgroups. There were no data relating to eosinophil subgroups in the FULFIL study.

Smoking status subgroup

For the smoking status subgroup, there was a strong trend for a higher reduction in the rate of moderate and severe exacerbations in ex-smokers compared to smokers for triple therapy comparisons with LAMA/ LABA in the largest study (IMPACT) and against LABA/ICS in TRILOGY. For the triple therapy comparisons with LAMA/LABA in TRIBUTE and LAMA in TRINITY, reduction in the rate of exacerbations was greater for smokers. There were no data relating to smoking status subgroups in the FULFIL or KRONOS studies.

Secondary outcomes

Time to first exacerbation

In the TRINITY, TRILOGY and IMPACT studies the time to first moderate or severe exacerbation was significantly extended with triple therapy compared with LAMA/LABA, LABA/ICS and LAMA (Fig. 4). In the KRONOS study, the time to first moderate or severe

			Rate ratio	95% Cl	Absolute difference	Pre- specified	Patient- based	Event- based
TRIBUTE (GLY/IND)		1			(70)"		ININ I "	NIN I "
Overall			0.85	0.72 to 0.995	15	Yes	~50	11
>1 exacerbation in the previous 12 months		·	- 0.96	0.75 to 1.25	4	Yes		NA
1 exacerbation in the previous 12 months		·	0.80	0.65 to 0.95	20	Yes		NA
Eosinophils ≥2%			0.81	0.66 to 0.98	19	Yes		NA
Eosinophils >0.2 x 10° cells/l			- 0.94 0.81	0.71 to 1.25	0 10	Yes		NA NA
Equinophilis $\leq 0.2 \times 10^{\circ}$ cells/L			0.87	0.69 to 1.01	13	Yes		NA
Ex-smokers			0.90	0.70 to 1.10 [¶]	10	Yes		NA
Current smoker			0.78	0.60 to 1.00¶	22	Yes		NA
IMPACT (UMEC/VI)								
Overall		HEH	0.75	0.70 to 0.81	25	Yes	~33	3
≥2 exacerbations in the previous 12 months			0.72	0.65 to 0.79	28	No		3
≤1 exacerbation in the previous 12 months			0.80	0.71 to 0.90	20	NO		5
Eosinophil count <150 cells/µL			0.66"	0.64 to 0.75" NA	32 12	No		3
Ex-smokers			0.70#	0.65 to 0.74#	30	No		3
Current smoker			0.86#	0.79 to 0.94#	14	No		7
KRONOS (GLY/FOR)⁺								
Overall			0.48	0.37 to 0.64	52	Yes	-	-
Eosinophil count ≥150 cells/mm ³			0.39	0.27 to 0.58	61	Yes		
Eosinophil count <150 cells/mm ³			0.61	0.41 to 0.90		Yes		
Overall			0.77	0.65 to 0.92	23	Yes	25	8
>1 exacerbation in the previous 12 months			0.67	0.48 to 0.94	33	No	20	3
1 exacerbation			0.83	0.67 to 1.02	17	No		14
Eosinophils ≥2%			0.77	0.61 to 0.95 [¶]	23	No		NA
Eosinophils <2%			0.79	0.60 to 1.10¶	21	No		NA
Eosinophils ≥0.2 × 10 ⁹ cells/L			0.80	0.65 to 0.99	20	No		NA
Eosinophils <0.2 × 10 [*] cells/L			0.72	0.55 to 1.00	28	No		NA
EX-SITIOREIS			0.07	0.55 to 0.85	33	NO		NA NA
IMPACT (VI/FF)			0.32	0.70 10 1.20-	0	NO		IN/A
Overall		H H H	0.85	0.80 to 0.90	15	Yes	~25	6
≥2 exacerbations in the previous 12 months		H H	0.89	0.82 to 0.97	11	No		8
≤1 exacerbation in the previous 12 months		H H	0.79	0.72 to 0.87	20	No		5
Eosinophil count ≥150 cells/µL			0.88#	NA	12	No		8
Eosinophil count <150 cells/µL			0.80*	NA	20	No		5
EX-SITUREIS Current smoker			0.64*	0.80 to 0.89"	16	No		6
FULFIL (FOR/B)*			0.04	0.70 10 0.01	10	NO		0
Overall			0.65	0.49 to 0.86	35	Yes	_	_
0/1 moderate exacerbations in the previous 12 mon	ths	· · · · · · · · · · · · · · · · · · ·	0.64	0.46 to 0.89	36	Yes		
≥2 moderate exacerbations in the previous 12 mont	hs	-	0.75	NA	25	Yes		
≥1 severe exacerbation in the previous 12 months	F		0.41	0.20 to 0.83	59	Yes		
Overall			0.82	0.58 to 1.17	18	Voc		
Eosinophil count ≥150 cells/mm ³			- 0.82	0.50 to 1.17	17	Yes	-	-
Eosinophil count <150 cells/mm ³		· · · · · · · · · · · · · · · · · · ·	⊣ 0.79	0.48 to 1.28	21	Yes		
Versus LAMA								
TRINITY (TIO)								
Overall			0.80	0.69 to 0.92	20	Yes	~25	9
>1 exacerbation in the previous 12 months			0.72	0.55 to 0.95	28	Yes		NA
r exacerbation			0.84	0.71 to 0.991	30	res		NA NA
Eosinophils <2%			0.70	0.75 to 1 17	7	Yes		NA
Eosinophils ≥0.2 × 10° cells/L			0.64	0.51 to 0.81	36	Yes		NA
Eosinophils <0.2 × 10° cells/L		-∔≡1	0.92	0.77 to 1.10	8	Yes		NA
Ex-smokers		⊧∎∳	0.85	0.70 to 1.03 [¶]	15	Yes		NA
Current smoker		►	0.76	0.63 to 0.94¶	22	Yes		NA
-								
	0.3	0.5 1.0	2.0					
	←	E	→					
		Favours triple Favo	ours comparators					

Fig. 3 Descriptive presentation of the annual rate of moderate and severe exacerbations for within-trial comparisons. *B* budesonide, *BDP* beclomethasone, *CI* confidence interval, *COPD* chronic obstructive pulmonary disease, *FF* fluticasone furoate, *FOR* formoterol fumarate, *FULFIL* Lung FUnction and quality of LiFe assessment in COPD with closed trlpLe therapy, *GLY* glycopyrronium bromide, *ICS* inhaled corticosteroids, *IMPACT* InforMing the PAthway of COPD Treatment, *IND* indacaterol, *LABA* long-acting β_2 agonist, *LAMA* long-acting muscarinic antagonist, *NA* not available, *NNT* number needed to treat, *TIO* tiotropium, *UMEC* umeclidinium, *VI* vilanterol. [#]Calculated. [¶]Estimated from graph. ⁺A 24-week study. Rate ratios, 95% CIs and NNT for annual rate of moderate and severe exacerbations between triple therapy and comparators, overall and by subgroups for each within-trial comparison of triple therapy and LAMA/LABA, LABA/ICS and LAMA. Cross-trial comparisons are descriptive only. NNT were only calculated for 52-week studies, ~ numbers used in calculation estimated from graph.

exacerbation was extended with triple therapy; the improvement was significant with LAMA/LABA (hazard ratio [HR]: 0.59; p < 0.0001 [Cox regression] and p = 0.0001 [log rank]), but not with LABA/ICS (HR: 0.75;

p = 0.0635 [Cox regression] and p = 0.0281 [log rank]). The TRIBUTE study showed a similar time to first moderate or severe exacerbation with triple therapy and LAMA/LABA (HR: 0.90; 95% CI: 0.76, 1.06; p = 0.22).



There were no data relating to time to first moderate or severe exacerbation in the FULFIL study.

Lung function

In the three studies that assessed the single-inhaler triple therapy GLY/FOR/BDP (TRINITY, TRILOGY and TRIBUTE), triple therapy resulted in a significant change from baseline in pre-dose FEV₁ compared with LAMA (61 mL; 95% CI: 37, 86; *p* < 0.0001 [52-week data]) [16], LABA/ICS (81 mL; 95% CI: 52, 109; *p* < 0.001 [26-week data], 63 mL; 95% CI: 32, 94; *p* < 0.001 [52-week data]) [17] and LAMA/LABA (22 mL overall; p < 0.05) [23]. In terms of FEV₁ change from baseline, patients were more likely to respond (defined as change from baseline in pre-dose $FEV_1 \ge 100 \text{ mL}$) to triple therapy than to LAMA (odds ratio [OR]: 1.61; 95% CI: 1.34, 1.93; p < 0.0001 [26-week data], 1.62; 95% CI: 1.35, 1.95; p <0.0001 [52-week data]) [16], LABA/ICS (OR: 2.30; 95% CI: 1.82, 2.91; *p* < 0.001 [26-week data], 2.06; 95% CI: 1.62, 2.62; p < 0.001 [52-week data]) [17] but not LAMA/LABA (OR: 1.19; 95% CI: 0.91, 1.55; p = 0.198]) [23]. A further post-hoc analysis of the TRINITY study considered different thresholds for FEV₁ response. At week 26, the proportion of responders (using a 50 mL threshold definition) was 48.0% for single-inhaler GLY/ FOR/BDP and 35.7% for TIO; using a 120 mL threshold the proportions were 36.7% for single-inhaler GLY/FOR/ BDP and 25.3% for TIO [32].

In the two studies that assessed the single-inhaler triple therapy UMEC/VI/FF (FULFIL and IMPACT), triple therapy resulted in a significant change from baseline in pre-dose FEV₁ compared with LABA/ICS (FOR/ B, 171 mL; 95% CI: 148, 194; p < 0.001 [24-week data], 179 mL; 95% CI: 131, 226; p < 0.001 [52-week data] [24]; VI/FF, 97 mL; 95% CI: 85, 109; p < 0.001 [52-week data]) [25] and LAMA/LABA (54 mL; 95% CI: 39, 69; p < 0.001[52-week data]) [25]. In terms of FEV₁ change from baseline, patients were more likely to respond (defined as change from baseline in trough FEV₁ ≥ 100 mL) to triple therapy than to LABA/ICS (FOR/B, OR: 4.03; 95% CI: 3.27, 4.97; p < 0.001 [24-week data], 4.79; 95% CI: 3.02, 7.61; p < 0.001 [52-week data]) [24]. FEV₁ response was not reported in the IMPACT study.

In the study that assessed the single-inhaler triple therapy GLY/FOR/B (KRONOS), triple therapy resulted in a significant change from baseline in pre-dose FEV₁ compared with LAMA/LABA (22 mL; 95% CI: 4, 39; p =0.0139), LABA/ICS (74 mL; 95% CI: 52, 95; p < 0.0001) [26]. FEV₁ response was not reported in the KRONOS study.

Quality of life

Across all the studies quality of life was assessed using the St. George's Respiratory Questionnaire (SGRQ) [33]. In the three studies that assessed the single-inhaler triple therapy GLY/FOR/BDP (TRINITY, TRILOGY and TRIBUTE), triple therapy resulted in greater improvement in SGRQ total score compared with TIO (at all timepoints except week 26 [weeks 4 and 12: p < 0.001; week 40: p < 0.01; week 52: p < 0.05]) [16], LABA/ICS (at weeks 4, 12 and 52 [week 52 mean treatment difference: -1.69; 95% CI: -3.20, -0.17; p = 0.029]) [17] and LAMA/LABA (week 4 and 12: $p \le 0.001$; week 26: p <0.05; week 40 and 52: p < 0.01) [23]. In terms of SGRQ response (defined as decrease from baseline in total score \geq 4 units), triple therapy was superior to TIO (OR: 1.32; 95% CI: 1.10, 1.57; *p* = 0.0024 [26-week data], 1.33; 95% CI: 1.11, 1.59; p = 0.0019 [52-week data]) [16] and LABA/ICS (OR: 1.52; 95% CI: 1.21, 1.91; p < 0.001 [26week data], 1.33; 95% CI: 1.06, 1.66; *p* = 0.014 [52-week

data]) [17], but not LAMA/LABA (OR: 1.22; 95% CI: 0.99, 1.51; *p* = 0.068 [52-week data]) [23].

In the two studies that assessed the single-inhaler triple therapy UMEC/VI/FF (FULFIL and IMPACT), triple therapy resulted in a greater improvement in SGRQ total score compared with LABA/ICS (FOR/B, at week 24 [difference: – 2.2 units; 95% CI: – 3.5, – 1.0; *p* < 0.001] [24]; VI/FF, at week 52 [difference: -1.8; 95% CI: -2.4, -1.1; p < 0.001]) [25] and LAMA/LABA (at week 52 [difference: -1.8; 95% CI: -2.6, -1.0; *p* < 0.001]) [25]. In the IMPACT study, there were similar changes in SGRQ total score for LAMA/LABA and LABA/ICS [25]. In terms of SGRQ response (defined as decrease from baseline in total score ≥ 4 units), triple therapy was superior to LABA/ICS (FOR/B, OR: 1.41; 95% CI: 1.16, 1.70; *p* < 0.001 [24-week data] [24]; VI/FF, OR: 1.41; 95% CI: 1.29, 1.55; *p* < 0.001 [52-week data]) [25] and LAMA/ LABA (OR: 1.41; 95% CI: 1.26, 1.57; p < 0.001 [52-week data]) [25].

In the study that assessed the single-inhaler triple therapy GLY/FOR/B (KRONOS), triple therapy resulted in nominally significant improvements in SGRQ total score over 24 weeks compared with LAMA/LABA (difference: -1.22; 95% CI: -2.30, -0.15; p = 0.0259), but not LAMA/ICS (difference: -0.45; 95% CI: -1.78, 0.87; p =0.5036) [26]. In terms of SGRQ response (defined as decrease from baseline in total score ≥ 4 units), triple therapy was associated with nominally significant improvements compared with LAMA/LABA (OR: 1.28; 95% CI: 1.01, 1.61; p = 0.0395) but not LABA/ICS (OR: 1.30; 95% CI: 0.97, 1.72; p = 0.0746) [26].

Safety

Across all the studies, SAEs were comparable between triple therapy and comparators. In the three studies that assessed the triple therapy GLY/FOR/BDP (TRINITY, TRILOGY and TRIBUTE), triple therapy was associated with a similar percentage of patients who experienced SAEs (13 [16]–15% [17, 23]) compared with TIO (15%) [16], LABA/ICS (18%) [17] and LAMA/LABA (17%) [23].

In the two studies that assessed the triple therapy UMEC/VI/FF (FULFIL and IMPACT), triple therapy was associated with a similar percentage of patients who experienced SAEs (5.4% [24] [24-week data], 10 [24]–22% [25] [52-week data]) compared with LABA/ICS (FOR/B, 5.7% [24-week data], 12.7% [52-week data] [24]; VI/FF, 21% [52-week data]) [25] and LAMA/LABA (23% [52-week data]) [25].

In the study that assessed the single-inhaler triple therapy GLY/FOR/B (KRONOS), triple therapy was associated with SAEs in 9% of patients compared with 11% for LAMA/LABA and 7% for LABA/ICS [26].

In all studies, except FULFIL and IMPACT, pneumonia was reported in 1 to 4% of patients, with similar incidences

in the treatment groups within each study. Pneumonia was defined as an adverse event (AE) by the investigator in each of these studies and additionally, in the KRONOS study, was validated using supporting diagnostic and treatment criteria.

In the FULFIL study, there was a higher rate of pneumonia (defined as an AE with supporting radiography) for triple therapy compared with LABA/ICS up to 24 weeks (2.1% versus 0.8%); the rates were similar between the two groups at 52 weeks (1.9% versus 1.8%). In the IMPACT study there was a higher incidence of pneumonia (defined as an AE of special interest with supporting radiography) with the triple therapy (UMEC/VI/FF; 8%) and LABA/ICS (7%), than with LAMA/LABA (5%). The rate of pneumonia per 1000 patient-years was 95.8 for triple therapy, 61.2 for LAMA/LABA and 96.6 for LABA/ICS. For pneumonia defined as a SAE with supporting radiography, the rate per 1000 patient-years was 53.3 for triple therapy, 32.4 for LAMA/LABA and 47.7 for LABA/ICS. The risk of pneumonia was significantly higher with triple therapy than with LAMA/LABA, as assessed in a time-to-first-event analysis (HR: 1.53; 95% CI: 1.22, 1.92; *p* < 0.001) [25].

In the TRINITY, TRILOGY, TRIBUTE and FULFIL studies, mortality was reported to be similar in each arm and not related to study medication. A pooled analysis of the time to death in the TRINITY, TRILOGY and TRIBUTE studies did not show a significant reduction in the risk of developing a fatal event compared with ICS-free treatments (LAMA and LAMA/LABA, HR: 0.72; 95% CI: 0.49, 1.06; *p* = 0.096) [22]. In the IMPACT study, for the prespecified endpoint, there was a statistically significant reduction in the risk of time to on- and off-treatment all-cause mortality for triple therapy compared to LAMA/LABA (HR: 0.71; 95% CI: 0.51, 0.99; unadjusted p = 0.043) but not LABA/ICS. Similar results were seen when off-treatment information was included. The percentage of on- and off-treatment deaths in each arm were 2.14% with UMEC/VI/FF, 2.35% with VI/FF and 2.90% with UMEC/VI [25]. In the KRONOS study, there were two deaths in the LAMA/LABA group that were considered to be related to study treatment [26].

Discussion

Overall, this systematic review demonstrates, that in patients with predominantly severe lung function impairment and a history of frequent exacerbations, singleinhaler triple therapy results in a reduction in moderate and severe exacerbations. Over 1 year, triple therapy reduced the frequency of moderate and severe exacerbations by 15 to 52% compared to LAMA/LABA, and 15 to 35% compared to LABA/ICS. The patient-based NNT ranged from around 25 to 50 (preventing one patient from having an event) and the event-based NNT ranged from around 3 to 11 (preventing one event). The absolute benefit appears to be greater in phenotypic subgroups with higher eosinophil counts or higher historical frequency of exacerbations and to some extent, exsmokers. The reductions in exacerbations come at the expense of a significant increase in pneumonia for some regimens, including pneumonia as a SAE.

Single-inhaler triple therapy showed some improvements in lung function and health-related quality of life over comparators. However, for the responder analysis (where threshold definitions reflected minimal clinicallyimportant differences), comparisons against LAMA/ LABA were not significant, indicating the differences were small and potentially not clinically relevant.

Previous reviews assessing triple therapy, in general, have demonstrated a reduction in exacerbations and an improvement in lung function and health-related quality of life but with an increased risk of pneumonia [3-7]. Our findings are consistent with a previously published review of multiple and single inhaler triple therapies, which included three of the six single inhaler clinical trials included in the present study. Although results for single inhalers were not reported separately, the results indicate that the benefit of triple therapy in reducing the risk of exacerbation is greater in patients with higher eosinophil counts [7]. Overall, the magnitude of clinical benefit for single-inhaler triple therapy, particularly across multiple subgroups, has not been well documented, which was the focus of our study. This study is the first to systematically review and descriptively assess the relative and absolute efficacy and safety of all six single-inhaler triple therapy studies compared to dual and monotherapy and explore differences in efficacy between phenotypic subgroups.

We also critically assessed the evidence base for singleinhaler triple therapy, which is currently not well documented. There were important differences in study designs and populations that would give rise to significant heterogeneity if class-level or drug-level meta-analyses of singleinhaler triple therapies were conducted, meaning results of such studies would need to be treated with caution. First, prior triple therapy was allowed in some trials and not others. The UMEC/VI/FF studies (FULFIL and IMPACT) allowed patients on triple therapy at the time of randomisation to enter the trial. In the IMPACT study 35% of patients in both the LAMA/LABA and LABA/ICS arms experienced ICS withdrawal at the time of randomisation. Second, some trials allowed abrupt withdrawal of ICS prior to study start. In the IMPACT study, more than 70% of patients randomised to LAMA/LABA were receiving ICS at randomisation. In the TRIBUTE study, all patients were switched to LAMA/LABA at the start of the 2-week run-in period. This included 65% of patients who were taking LABA/ICS or LAMA/ICS for at least 2 months before Page 10 of 13

study entry. No subgroup analyses were done according to previous treatment. Third, there were differences in the severity of COPD between the trials, particularly in relation to prior exacerbation history, with one study (KRONOS) including patients without a prior history of exacerbations, a group not recommended for treatment with triple therapy according to the recent GOLD update [2]. Fourth, different regimens were used, which may impact the risks and benefits observed. Finally, the IMPACT trial also allowed patients with a prior history of asthma. As these data were not recorded, however, it is difficult to know whether this influenced results. All of these factors could lead to exaggeration of the benefits of ICS. Some studies have shown that abrupt withdrawal of ICS may lead to an increase in exacerbations in some patients [34-36], unlike a managed stepwise withdrawal of ICS [37]. Class-level meta-analyses including free-combination triple therapies and limited data on single-inhaler triple therapies have found statistical and clinical heterogeneity to be a significant concern [6, 7].

The IMPACT study reported a benefit in the risk of time to all-cause mortality for triple therapy compared to LAMA/LABA. As none of the single-inhaler triple therapy studies were powered for a mortality outcome, these results should be treated with caution and investigated further. Two prospective, adequately-powered, large-scale studies with a primary outcome of mortality failed to demonstrate a significant reduction in mortality associated with the use of ICS-containing regimens [38, 39].

Our review attempted to determine whether the benefits associated with triple therapy in COPD were clinically meaningful by assessing and comparing relative and absolute improvements in outcomes as well as the NNT. These latter results are of interest as they translate the results of randomised trials into a value that clinicians can use in decision making. Focusing on the 12-month trials, we found relatively high NNTs for therapy at 12 months (> 20) in terms of reducing the numbers of individuals experiencing exacerbations but lower NNTs (3 to 11) for reducing one exacerbation per patient per year. The discrepancy between these two numbers suggests that the impact of triple therapy is not necessarily to increase the number of patients staying exacerbation free, but that the largest effect is in reducing the number of events in patients having multiple events. Such patients are most likely the high-risk individuals with a higher rate of baseline exacerbations and blood eosinophilia as suggested by the subgroup data. The evidence therefore supports a targeted approach to triple therapy use in those patients at highest risk of exacerbation with higher levels of blood eosinophilia. It also implies that the vast majority of patients with a lower baseline risk of exacerbation and lower blood eosinophilia may experience less benefit from ICS, as reflected in the GOLD update [2]. The increased risk of ICS, particularly in

relation to pneumonia, is well documented [40], and likely to vary by treatment-related factors (e.g. regimen and dose) and study design-related factors (e.g. patient characteristics and definitions used) [41]. In the present study, the pneumonia risk for single-inhaler triple therapy ranged from no reported increase to a significant increase. Taking the study demonstrating the highest risk and translating it into an event-based number needed to harm (NNH) to induce pneumonia in at least one patient over one year, gives an NNH of 33.

The next step is to more clearly define patient groups where ICS are of benefit. This review demonstrated that patients with higher eosinophil counts, higher prior exacerbation frequency and ex-smokers, in general, benefited more from single-inhaler triple therapy; however, results for all three phenotypes were mixed. This may have resulted, in part, from the thresholds that were used in these studies. The GOLD 2019 document suggests that patients with blood eosinophil counts < 100 and \geq 300 cells/µL point to subgroups with lower and higher likelihood of treatment benefit with ICS [2]. There is a need to validate and refine this threshold and identify other patient characteristics that can support the treatment decision to step up to triple therapy.

The limitations of our systematic review largely reflect the shortcomings of the included studies. First, there was limited information reported in the studies on subgroups and information needed to calculate patient-based NNTs therefore a number of the results were either calculated based on available data or estimated from forest plots or Kaplan-Meier curves. Second, interpretation of subgroup data needs to be treated with caution. Many subgroup analyses were not prespecified in the original study and studies were not powered to demonstrate an interaction effect. In addition, although the inclusion criteria for this review were carefully chosen to answer the research question, broadening the criteria, for example by including all types of randomised controlled trials of any duration, may have had an impact on the interpretation of the results. However, a review with broader inclusion criteria, presented similar overall findings [7].

Overall the evidence from these studies support the GOLD 2019 treatment strategy in that triple therapy should be reserved for a select group of patients with COPD with frequent exacerbations and higher eosino-phil counts thereby maximising benefit and minimising risk [2]. Future research on specific patient phenotype thresholds that can support treatment and funding decisions is now required from well-designed, robust, clinical trials.

Conclusions

Our analysis of single-inhaler triple therapy trials suggests that in the patient populations included in these studies, 25–50 patients would need to be treated with single-inhaler triple therapy to prevent one patient from having a moderate or severe exacerbation over one year compared with dual therapy, with an NNH for the pneumonia AE of around 33 and upwards. This emphasises that we must weigh benefits and risks of ICS in individual patients. Our analyses suggest benefit appears to be greatest in patients with higher eosinophil counts, greater historical frequency of exacerbations, and exsmokers. This must be balanced against the increased risk for AEs including pneumonia.

Supplementary information

Supplementary information accompanies this paper at https://doi.org/10. 1186/s12931-019-1213-9.

Additional file 1: Table S1. Eligibility criteria. Table S2. Search strategy. Table S3. Overview of excluded studies during the full-text analysis stage in alphabetical order. Figure S1. Risk of bias assessment summary.

Abbreviations

AE: Adverse event; B: Budesonide; BDP: Beclomethasone; CAT: COPD assessment test; CI: Confidence interval; COPD: Chronic obstructive pulmonary disease; FEV₁: Forced expiratory volume in one second; FF: Fluticasone furoate; FOR: Formoterol fumarate; FULFIL: Lung FUnction and quality of LiFe assessment in COPD with closed trlpLe therapy; GLY: Glycopyrronium bromide; GOLD: Global Initiative for Chronic Obstructive Lung Disease; HR: Hazard ratio; ICS: Inhaled corticosteroids; IMPACT: InforMing the PAthway of COPD Treatment; IND: Indacaterol; LABA: Long-acting β_2 agonist; LAMA: Long-acting muscarinic antagonist; MF: Mometasone furoate; NNH: Number needed to harm; NNT: Number needed to treat; NR: Not reported; OR: Odds ratio; PRISMA: Preferred Reporting Items for Systematic Reviews and Meta-Analyses; RR: Rate ratio; SAE: Serious adverse event; SD: Standard deviation; SGRQ: St. George's Respiratory Questionnaire; TIO: Tiotropium; UMEC: Umeclidinium; VI: Vilanterol

Acknowledgements

Not applicable.

Authors' contributions

SL and JLa conceived and started the study. SL coordinated and supervised the study. NP, JLa and SL designed the study. JLe, NP, SL and JLa performed the study search, extracted and interpreted the data. NP and NE performed the analyses. SL and JLe wrote the first draft of the manuscript. MKH and JDC provided interpretation of the data and critically revised the manuscript for important intellectual content. All authors reviewed and provided critical comments on drafts. All authors have approved the submitted version. All authors can take responsibility for the integrity of the data and the accuracy of the data analysis. The corresponding author attests that all listed authors meet authorship criteria and that no others meeting the criteria have been omitted.

Funding

The current analysis was supported by funding from Boehringer Ingelheim. The funder of the study had no role in its design or conduct, in the collection, analysis, or interpretation of the data, or in the writing of the manuscript.

Availability of data and materials

All data generated or analysed during this study are included in this published article [and its supplementary information files].

Ethics approval and consent to participate

Not applicable.

Consent for publication

Not applicable.

Competing interests

SL, JLe, NP, NE, and JLa report personal fees from Boehringer Ingelheim during the conduct of the study and outside the submitted work. MKH reports consulting for Boehringer Ingelheim, GlaxoSmithKline, AstraZeneca and Mylan as well as research support from Novartis and Sunovion. JDC reports grants and personal fees from Boehringer Ingelheim, GlaxoSmithKline, AstraZeneca, Grifols and Gilead Sciences outside the submitted work. The authors have no financial relationships with any organisations, or other relationships or activities, that might have influenced the submitted work in the previous three years.

Author details

¹Maverex Limited, Manchester, UK. ²University of Michigan, Women's Respiratory Clinic, Ann Arbor, USA. ³University of Dundee and Ninewells Hospital and Medical School, Dundee, UK.

Received: 21 March 2019 Accepted: 11 October 2019 Published online: 04 November 2019

References

- GBD 2015 Chronic Respiratory Disease Collaborators. Global, regional, and national deaths, prevalence, disability-adjusted life years, and years lived with disability for chronic obstructive pulmonary disease and asthma, 1990– 2015: a systematic analysis for the Global Burden of Disease Study 2015. Lancet Respir Med. 2017;5:691–706.
- GOLD. Global strategy for the diagnosis, management, and prevention of chronic obstructive pulmonary disease [Internet]. 2019. Available from: https://goldcopd.org/wp-content/uploads/2018/11/GOLD-2019-v1.5-FINAL-04Nov2018_WMS.pdf
- Rojas-Reyes MX, Garcia Morales OM, Dennis RJ, Karner C. Combination inhaled steroid and long-acting beta(2)-agonist in addition to tiotropium versus tiotropium or combination alone for chronic obstructive pulmonary disease. Cochrane Database Syst Rev. 2016; CD008532.
- Liu Y, Shi H, Sun X, Zhang D, Zhang Y, Yang K, et al. Benefits of adding fluticasone propionate/salmeterol to tiotropium in COPD: a meta-analysis. Eur J Intern Med. 2014;25:491–5.
- Kwak MS, Kim E, Jang EJ, Kim HJ, Lee CH. The efficacy and safety of triple inhaled treatment in patients with chronic obstructive pulmonary disease: a systematic review and meta-analysis using Bayesian methods. Int J COPD. 2015;10:2365–76.
- Zheng Y, Zhu J, Liu Y, Lai W, Lin C, Qiu K, et al. Triple therapy in the management of chronic obstructive pulmonary disease: systematic review and meta-analysis. BMJ. 2018;363:k4388.
- Cazzola M, Rogliani P, Calzetta L, Matera MG. Triple therapy versus single and dual long-acting bronchodilator therapy in chronic obstructive pulmonary disease: a systematic review and meta-analysis. Eur Respir J. 2018;52:1801586.
- Brusselle G, Price D, Gruffydd-Jones K, Miravitlles M, Keininger DL, Stewart R, et al. The inevitable drift to triple therapy in COPD: an analysis of prescribing pathways in the UK. Int J Chron Obs Pulmon Dis. 2015;10:2207–17.
- Lane DC, Stemkowski S, Stanford RH, Tao Z. Initiation of triple therapy with multiple inhalers in chronic obstructive pulmonary disease: an analysis of treatment patterns from a U.S. retrospective database study. J Manag care spec pharm. United States. 2018;24:1165–72.
- Chalmers JD, Tebboth A, Gayle A, Ternouth A, Ramscar N. Determinants of initial inhaled corticosteroid use in patients with GOLD a/B COPD: a retrospective study of UK general practice. NPJ Prim Care Respir Med. 2017;27:43.
- Vanfleteren LEGW, Ullman A, Nordenson A, Andersson A, Andelid K, Fabbri LM. Triple therapy (ICS/LABA/LAMA) in COPD: thinking out of the box. ERJ Open Res. 2019;5:00185–2018.
- 12. Hutton B, Salanti G, Caldwell DM, Chaimani A, Schmid CH, Cameron C, et al. The PRISMA extension statement for reporting of systematic reviews incorporating network meta-analyses of health care interventions: checklist and explanations. Ann Intern Med. 2015;162:777.
- Langham S, Langham J, Pooley N. Efficacy and safety of fixed-dose combination triple therapy in comparison to treatment with long-acting muscarinic antagonists (LAMAs), long-acting ß2 agonists (LABAs). LABA/

LAMA or LABA/inhaled corticosteroids (ICS) for the treatments of chronic obstruc Available from: www.crd.york.ac.uk/PROSPERO/display_record. php?ID=CRD42018102125. .

- European Medicines Agency. Guideline on clinical investigation of medicinal products in the treatment of chronic obstructive pulmonary disease (COPD). Available from: https://www.ema.europa.eu/documents/scientific-guideline/ guideline-clinical-investigation-medicinal-products-treatment-chronicobstructive-pulmonary-disease_en.pdf. Accessed 23 Oct 2018.
- 15. Higgins J, Green S. Cochrane Handbook for Systematic Reviews of Interventions Version 5.1.0. Available from: Handbook.cochrane.org/.
- Vestbo J, Papi A, Corradi M, Blazhko V, Montagna I, Francisco C, et al. Single inhaler extrafine triple therapy versus long-acting muscarinic antagonist therapy for chronic obstructive pulmonary disease (TRINITY): a double-blind, parallel group, randomised controlled trial. Lancet. 2017;389:1919–29.
- Singh D, Papi A, Corradi M, Pavlišová I, Montagna I, Francisco C, et al. Single inhaler triple therapy versus inhaled corticosteroid plus long-acting β2-agonist therapy for chronic obstructive pulmonary disease (TRILOGY): a double-blind, parallel group, randomised controlled trial. Lancet. 2016;388:963–73.
- Tabberer M, Lomas DA, Birk R, Brealey N, Zhu C-Q, Pascoe S, et al. Oncedaily triple therapy in patients with COPD: patient-reported symptoms and quality of life. Adv Ther. 2018;35:56–71.
- Fabbri LM, Rabe KF, Martinez FJ, Ferguson GT, Wang C, Ichinose M, et al. PA688 Impact of eosinophil levels on lung function and exacerbation benefits with co-suspension delivery technology budesonide/ glycopyrronium/formoterol metered dose inhaler (BGF MDI) in COPD (KRONOS study subgroup analysis). Presented at: ERS International Congress; Paris, France; 15–19 September; 2018.
- Halpin DMG, Birk R, Brealey N, Criner GJ, Dransfield MT, Hilton E, et al. Single-inhaler triple therapy in symptomatic COPD patients: FULFIL subgroup analyses. ERJ Open Res. 2018;4:00119–2017.
- Singh D, Barnhart F, Brealey N, Criner G, Dransfield M, Halpin D, et al. PA4383 FF/UMEC/VI versus FF/VI and UMEC/VI in the ELLIPTA device in patients with COPD: Subgroup analyses from the IMPACT trial. Presented at: ERS International Congress; Paris, France; 15–19 September; 2018.
- Scuri M, Fabbri LM, Singh D, Roche N, Corradi M, Alessandro G, et al. Reduction in fatal events with ICS-containing medications: results of safety pooled analysis from the TRILOGY. TRINITY and TRIBUTE studies Am J Respir Crit Care Med. 2018;197:A7725.
- Papi A, Vestbo J, Fabbri L, Corradi M, Prunier H, Cohuet G, et al. Extrafine inhaled triple therapy versus dual bronchodilator therapy in chronic obstructive pulmonary disease (TRIBUTE): a double-blind, parallel group, randomised controlled trial. Lancet. 2018;391:1076–84.
- Lipson DA, Barnacle H, Birk R, Brealey N, Locantore N, Lomas DA, et al. FULFIL trial: once-daily triple therapy in patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 2017;196:438–46.
- Lipson DA, Barnhart F, Brealey N, Brooks J, Criner GJ, Day NC, et al. Oncedaily single-inhaler triple versus dual therapy in patients with COPD. N Engl J Med. 2018;378:1671–80.
- Ferguson GT, Rabe KF, Martinez FJ, Fabbri LM, Wang C, Ichinose M, et al. Triple therapy with budesonide/glycopyrrolate/formoterol fumarate with cosuspension delivery technology versus dual therapies in chronic obstructive pulmonary disease (KRONOS): a double-blind, parallel-group, multicentre, phase 3 randomised controlled tr. Lancet Respir Med. 2018;6:747–58.
- Scuri M, Vestbo J, Papi A, Corradi M, Spinola M, Montagna I, et al. Association of incident pneumonia and exacerbations with extrafine triple therapy in one single inhaler in COPD patients: a post-hoc analysis from TRILOGY and TRINITY studies. Thorax. 2017;72:A231–2.
- Scuri M, Spinola M, Vezzoli S, Muraro A, Petruzzelli S, Vestbo J. Efficacy of CHF5993, a novel triple extrafine combination treatment in COPD patients at high risk of exacerbations: a sub-group analysis of the TRINITY study. Am J Respir Crit Care Med. 2017;195:A7710.
- Hilton E, Brearley N, Birk R, Zhu C-Q. P272 improvements in exacerbation rates with single inhaler triple therapy versus dual ics/laba therapy in patients with advanced chronic obstructive pulmonary disease (COPD): subgroup analyses of the phase III FULFIL study. Thorax. 2017;72:A231.
- Tabberer M, Lomas D, Barnacle H, Birk R, Brealey N, Zhu C-Q, et al. Single inhaler, once-daily, triple therapy (ICS/LAMA/LABA) in patients with advanced COPD: health-related quality of life (HRQoL) and activity outcome results from the FULFIL trial. Am J Respir Crit Care Med. 2017;195:A6742.
- 31. GlaxoSmithKline. A study comparing the efficacy, safety and tolerability of fixed dose combination (FDC) of FF/UMEC/VI with the FDC of FF/VI and

UMEC/VI; administered once-daily via a dry powder inhaler (DPI) in subjects with chronic obstructive pulmonary disease (COPD). Available from: Clinicaltrials.gov/ct2/show/NCT02164513.

- Scuri M, Vestbo J, Papi A, Corradi M, Montagna I, Francisco C, et al. Effect of extrafine single inhaler triple therapy on lung function and use of rescue medication: results from the TRINITY study. Thorax. 2017;72:A230.
- Jones PW, Quirk FH, Baveystock CM, Littlejohns P. A self-complete measure of health status for chronic airflow limitation. The St. George's respiratory questionnaire. Am Rev Respir Dis. 1992;145:1321–7.
- Vestbo J, Anderson JA, Calverley PMA, Celli B, Ferguson GT, Jenkins C, et al. Bias due to withdrawal in long-term randomised trials in COPD: evidence from the TORCH study. Clin Respir J. 2011;5:44–9.
- Kaplan AG. Applying the wisdom of stepping down inhaled corticosteroids in patients with COPD: a proposed algorithm for clinical practice. Int J Chron Obstruct Pulmon Dis. 2015;10:2535–48.
- 36. Suissa S, Drazen JM. Making sense of triple inhaled therapy for COPD. N Engl J Med. 2018;378:1723–4.
- Magnussen H, Disse B, Rodriguez-Roisin R, Kirsten A, Watz H, Tetzlaff K, et al. Withdrawal of inhaled glucocorticoids and exacerbations of COPD. New Engl J Med. 2014;371:1285–94.
- Vestbo J, Anderson JA, Brook RD, Calverley PMA, Celli BR, Crim C, et al. Fluticasone furoate and vilanterol and survival in chronic obstructive pulmonary disease with heightened cardiovascular risk (SUMMIT): a doubleblind randomised controlled trial. Lancet. 2016;387:1817–26.
- Calverley PMA, Anderson JA, Celli B, Ferguson GT, Jenkins C, Jones PW, et al. Salmeterol and fluticasone propionate and survival in chronic obstructive pulmonary disease. New Engl J Med. 2007;356:775–89.
- Kew KM, Seniukovich A. Inhaled steroids and risk of pneumonia for chronic obstructive pulmonary disease. Cochrane Database Syst Rev. 2014; CD010115.
- Tashkin DP, Strange C. Inhaled corticosteroids for chronic obstructive pulmonary disease: what is their role in therapy? Int J Chron Obstruct Pulmon Dis. 2018;13:2587–601.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

